Author's response to reviews

Title: Advanced Age, Altered Level of Consciousness and a New Diagnosis of Diabetes are Independently Associated with Hypernatremia in Hyperglycaemic Crisis

Authors:

Chukwuma Ekpebegh (chuksekpebegh@yahoo.com)
Benjamin Longo-Mbenza (longombenza@gmail.com)
Nge Okwe (papy_nge@yahoo.fr)
A Ogbera (chuksekpebegh@yahoo.com)
N Tonjeni (nttonjeni@yahoo.com)

Version: 3 Date: 7 April 2011

Author's response to reviews:

Major revision
1. Author list is arranged according the preferred style
2. Table 4 is now cited in the text
3. The colour of the tables is corrected

Minor revisions
1. We include email of all authors in the title page
2. All highlighting are removed